Author's response to reviews

Title: Characteristics of attempted self-poisoning in Sri Lanka; a systematic review

Authors:

Thilini Rajapakse Dr (gemba471@gmail.com)
Kathy Griffiths Prof (kathy.griffiths@anu.edu.au)
Christensen Helen Prof (h.christensen@blackdog.org.au)

Version: 2 Date: 21 October 2012

Author's response to reviews:

The Editors
Dear Sir/ Madam

During the initial submission of this article, by oversight the following were not uploaded: competing interests, authors' contributions, authors' information and acknowledgements.

We have now revised the submission to include the following documents (which have been uploaded separately from the main text of the article).

The main text of the article was not changed during this revision.

We apologize for any inconvenience caused,

Thank you,

Yours sincerely

Thilini Rajapakse
Corresponding author